Author’s response to reviews

Title: Identifying the primary outcome for a randomised controlled trial in rheumatoid arthritis: the role of a discrete choice experiment

Authors:

Eugena Stamuli (eugena.stamuli@york.ac.uk)
David Torgerson (david.torgerson@york.ac.uk)
Matthew Northgraves (matt.northgraves@york.ac.uk)
Sarah Ronaldson (sarah.ronaldson@york.ac.uk)
Lindsey Cherry (l.cherry@soton.ac.uk)

Version: 2 Date: 21 Nov 2017

Author’s response to reviews:

Dear Reviewer,

Thank you for your comments. Please find below the point-by-point response to your comments

1. Do not use all capital letters for subheadings in abstract
I have now changed all the subheadings to sentence case

2. Use (i), (ii), (iii), etc for lists rather than dot points
I have now changed the dot points to (i), (ii), (iii)

3. Move tables to the end of the document
I have now moved tables to the end of the document

4. Table 3: insert spaces before bracketed standard errors
I have now inserted spaces before bracketed SE
5. Change subheading “Discussion and Conclusions” to “Discussion”

I have changed subheading to "Discussion"

6. Abbreviate all journal names in reference list using correct PubMed abbreviations

I have abbreviated all Journal names using PubMed abbreviations

Kind regards

Eugena Stamuli